



**Baishideng
Publishing
Group**

7041 Koll Center Parkway, Suite
160, Pleasanton, CA 94566, USA
Telephone: +1-925-399-1568
E-mail: bpgoffice@wjgnet.com
https://www.wjgnet.com

Name of journal: World Journal of Hepatology

Manuscript NO: 67493

Title: Video-illustrated case report of a focal nodular hyperplasia associated with a giant hepatocellular adenoma: management and review of the literature

Reviewer's code: 05429684

SPECIFIC COMMENTS TO AUTHORS

I really enjoyed participating in the review of this manuscript and I want to congratulate the authors for their excellent work. The case report was described in an objective and concise manner. The discussion excluded the doubts that could arise regarding the treatment of different liver injuries, mentioned here.

Point by point response

Comment: Introduction line 5

- Reviewer: «You could better explain to the lay reader what the multinodular pathway is.»
 - **Response:** Thank you for pointing this error of typo. We changed it with the correct word, giving a more logical sense to our description.

No other comments were made. Thank you